Author's response to reviews

Title: The use of chemotherapy regimens carrying a moderate or high risk of febrile neutropenia and the corresponding management of febrile neutropenia: an expert survey in breast cancer and non-Hodgkin’s lymphoma

Authors:

Laetitia Gerlier (LGerlier@be.imshealth.com)
Mark Lamotte (MLamotte@be.imshealth.com)
Ahmad Awada (ahmad.awada@bordet.be)
André Bosly (andre.bosly@uclouvain.be)
Greet Bries (greet.bries@sezkturnhout.be)
Véronique Cocquyt (veronique.cocquyt@ugent.be)
Christian Focan (christian.focan@chc.be)
Stéphanie Henry (stephanie.henry@cmsenamur.be)
Yassine Lalami (yassine.lalami@bordet.be)
Jean-Pascal Machiels (jean-pascal.machiels@onco.ucl.ac.be)
Jeroen Mebis (jeroen.mebis@virgajesse.be)
Nicole Straetmans (nicole.straetmans@scarlet.be)
Didier Verhoeven (didier.verhoeven@klina.be)
Luc Somers (luc.somers@oncologx.be)

Version: 5 Date: 15 November 2010

Author's response to reviews:

Please find the revised version including additional editorial comments regarding the author names and order, as well as the list of abbreviations in correct format.

Sorry for these last style discrepancies, I hope the paper looks ‘fully compliant’ with BMC Cancer instructions now.

Best regards
Laetitia.